Author's response to reviews

Title: Embedding effective depression care: using theory for primary care organisational and systems change

Authors:

Jane M Gunn (j.gunn@unimelb.edu.au)
Victoria J Palmer (vpalmer@unimelb.edu.au)
Christopher F Dowrick (cfd@liverpool.ac.uk)
Helen E Herrman (h.herrman@unimelb.edu.au)
Frances E Griffiths (F.E.Griffiths@warwick.ac.uk)
Renata Kokanovic (renata.kokanovic@arts.monash.edu.au)
Grant A Blashki (gblashki@unimelb.edu.au)
Kelsey L Hegarty (k.hegarty@unimelb.edu.au)
Caroline L Johnson (c.johnson@unimelb.edu.au)
Maria Potiriadis (m.potiriadis@unimelb.edu.au)
Carl R May (c.r.may@newcastle.ac.uk)

Version: 5 Date: 29 July 2010

Author's response to reviews:

29/07/2010

Dear Editor,

Thank you for the email regarding formatting changes that was required of our manuscript. The revisions outlined in the email have all been addressed and the manuscript is ready for uploading.

Yours sincerely,

Professor Jane Gunn